\documentclass{acm_proc_article-sp}

\usepackage{subfigure}

\bibliographystyle{plain} 


% to latex the file type 
% latex sample.tex
% to view to file:
% xdvi sample.dvi

% to include the references:
% bibtex sample
% latex sample.tex
%latex sample.tex



\begin{document}


\title{No Title Yet}

\numberofauthors{1}
\author{
%
% The command \alignauthor (no curly braces needed) should
% precede each author name, affiliation/snail-mail address and
% e-mail address. Additionally, tag each line of
% affiliation/address with \affaddr, and tag the
%% e-mail address with \email.
\alignauthor Fill in Name Here\\
       \affaddr{Computer and Information Sciences}\\ \affaddr{
       University of Delaware}\\ \affaddr{Newark, DE 19716}\\
       \email{fill-in@cis.udel.edu}
}
\maketitle

\begin{abstract}
This is the abstract, usually 300 words or less.

\end{abstract}


\section{Introduction}
Motivation for overall topic area.
One paragraph stating state of the art
One paragraph stating current restrictions, problems with
existing work in the area.
Goals of this proposed research.
Overall approach and main tasks of this proposed research to accomplish goals.
How the proposal is organized.

\section{Background}
\section{Open Problems }
\subsection{Problem one here}
\subsection{Problem two here}
\section{Proposed Approach(es)}
\section{Related Work}

\section{Conclusions and Future Work}

Here is an example citation~\cite{Pollock,family,Java,ASUbook,holley:tse:81,simple}.


\bibliography{sample}  % includes sample.bib
 
\end{document}

